Author's response to reviews

Title: Prevalence and profile of unrecognised palliative patients in hospitals

Authors:

Marianne S Desmedt (marianne.desmedt@uclouvain.be)
Yolande L de la Kethulle (yolande.delakethulle@uclouvain.be)
Myriam I Deveugele (myriam.deveugele@ugent.be)
Emmanuel A Keirse (emmanuel.keirse@med.kuleuven.be)
Dominique J Paulus (dominique.paulus@kce.fgov.be)
Johan J Menten (johan.menten@uz.kuleuven.be)
Steven R Simoens (steven.simoens@pharm.kuleuven.be)
Paul J Vanden Berghe (paul.vanden.berghe@palliatief.be)
Claire M Beguin (claire.beguin@uclouvain.be)

Version: 6 Date: 28 February 2011

Author's response to reviews: see over
Answers to the Editorial Production Team

Thank you very much for your help.

Major changes

*Title: please ensure that the title is identical in the manuscript and the submission system.*

I have corrected it. Normally, they are the same now. In case of problem, the title of the paper is the correct one.

*Abstract: please ensure that the Abstract present in the manuscript matches exactly what is in the submission system.*

I have verified, the two abstracts are identical.

*Additional files: all additional files must be mentioned in the text ….*

The additional files are mentioned in the 9th paragraph of the section ‘Method’ (I have added a sentence). A section additional file is mentioned at the end of the paper.

Minor changes

*All text in black:* It has been done.

*Title:* The typography has been adapted as you asked.

*Figures:* I have tried to crop the figure. I hope that now it is in order for you.

*Tables … vertical lines…:* they are removed

*Additional files:* the ethics approval are removed